Author's response to reviews

Title: Cholestatic Jaundice, Acute Renal Failure and Acute Pancreatitis Secondary to Methandrostenolone: a Case Report of Recreational Anabolic Steroid Use

Authors:

Greg Rosenfeld (grosenfeld@telus.net)
Albert Chang (albert.chang@utoronto.ca)
Michael Poulin (michael.poulin@vch.ca)
Peter Kwan (pwkwan@gmail.com)
Eric Yoshida (eric.yoshida@vch.ca)

Version: 4 Date: 27 October 2010

Author's response to reviews: see over
Dear Editor,

Thank you for reviewing this manuscript. Please note the attached changes in the document.

Referee 1:

1. The papers reporting the use of AAS enriched food supplements and renal failure were reviewed and incorporated into the manuscript with additional references added.
2. The Hepatology paper concerning causality for DILI was incorporated into the document and the reference added.
3. The patient’s age was added
4. Transaminase has been replaced by aminotransferase throughout
5. The Introduction was shortened
6. The athlete’s names were removed

Referee 2:

1. The Title has been changed
2. The Abstract was shortened and modified as suggested
3. The Introduction was modified as suggested
4. A Table with all of the pertinent lab values has been added
5. Discussion has been modified and the pertinent paper discussing AKI has been incorporated.

Thank you for reviewing these changes and I hope that you will find that the paper now is sufficient for publication in your journal.

Sincerely,

Greg Rosenfeld